**Author’s response to reviews**

**Title:** Exploring computerised cognitive training as a therapeutic intervention for people with Huntington's disease (CogTrainHD): protocol for a randomised feasibility study

**Authors:**

Emma Yhnell (YhnellE@cardiff.ac.uk)

Hannah Furby (FurbyH1@cardiff.ac.uk)

Rachel Breen (BreenR@cardiff.ac.uk)

Lucy Brookes-Howell (Brookes-HowellLC@cardiff.ac.uk)

Cheney Drew (DrewC5@cardiff.ac.uk)

Rebecca Playle (PlayleRA@cardiff.ac.uk)

Gareth Watson (WatsonGL@cardiff.ac.uk)

Claudia Metzler-Baddeley (Metzler-BaddeleyC@cardiff.ac.uk)

Anne Rosser (RosserAE@cardiff.ac.uk)

Monica Busse (BusseME@cardiff.ac.uk)

**Version:** 1  **Date:** 09 Jan 2018

**Author’s response to reviews:**

Dear Editor,

We would like to thank both yourself and the reviewer for your comments on our submission, which will help to improve our manuscript. We wish to address the specific points below.

Reviewer reports:

Reviewer #1: A very interesting protocol for a study in an interesting and important area.

General points

Has fatigue been considered as a con-founder in the study?
Response:

Fatigue has been considered as a potential barrier to the completion of the brain training. Alongside apathy, this is something that we can explore as part of the semi-structured interviews, particularly at the outcome interview where we ask participants about barriers and their ability to complete the brain training.

Specific Points:

1. Page 9 line 214, how was the protocol of 3 times per week for twelve weeks determined

Response:

The protocol was determined based on the recommendations of the software provider, the text has now been updated to include this information.

Editorial comments:

Please correct typos along the text.

Response:

The manuscript has been proof read and typographical errors have been corrected as necessary throughout.

Yours sincerely,

Dr. Emma Yhnell B.Sc. Ph.D. FHEA